Author’s response to reviews

Title: Course of Post Traumatic Stress Disorder and healthcare utilisation among resettled refugees in the Netherlands

Authors:

Majda Lamkaddem (m.lamkaddem@amc.uva.nl)
Karien Stronks (k.stronks@amc.uva.nl)
Walter D Devillé (w.deville@nivel.nl)
Miranda Olff (m.olff@amc.uva.nl)
Annette A.M. Gerritsen (annette.gerritsen@epiresult.com)
Marie-Louise Essink-Bot (m.l.essink-bot@amc.uva.nl)

Version: 2 Date: 21 June 2013

Author’s response to reviews:

Dear Sir, Madam,

Following your comments, an ethics statement was added to the method section of this revised version of the manuscript.

Sincerely,

Majda Lamkaddem